Author’s response to reviews

Title: Establishment and utility assessment of the Posterior Reversible Encephalopathy Syndrome early warning score (PEWS) scale

Authors:

Li-Ping Zou (zouliping21@sina.com)
Hui Li (lihuheihei@163.com)
Li-Ying Liu (liuliying0812@hotmail.com)
Yang-Yang Wang (schopenhauersays@gmail.com)
Ying Liu (liuyingdr@163.com)
Jing Chen (chenjing@scmc.com.cn)
Lin-Yan Hu (huli3188@aliyun.com)
Meng-Jia Liu (liumengjiawang@126.com)
Meng-Na Zhang (zhangmengna2014@sina.com)
Qian Lu (drluqian@163.com)
Shu-Fang Ma (mashufang1990@sina.com)

Version: 5 Date: 19 Sep 2018

Author’s response to reviews:

Dear Editor DR. Naoum P Issa:

Thank you for concerning our manuscript entitled “Establishment and utility assessment of the Posterior Reversible Encephalopathy Syndrome early warning score (PEWS) scale” (manuscript number: NURL-D-18-00117R4). Thank you for your helpful advice. Below are our responses to the two comments.

Editor Comments:

Q1: Are these PRES patients or NON-PRES patients?
Response: Thank you for your suggestion. It has been modified as “15 NON-PRES cases with a median age of 9.5 years old” in page 11.

Q2: Should this be: “Of the three other patients, one patient with a PEWS score of 10 points presented no significant imaging abnormalities.”

Response: Thank you for your suggestion. We are so sorry for our unclear expression. We have changed the sentence in page 13.